Author's response to reviews

**Title:** Pilot Randomized Trial of Therapeutic Hypothermia with Serial Cranial Ultrasound and 18-22 Month Follow-up for Neonatal Encephalopathy in a Low Resource Hospital Setting in Uganda: Study Protocol

**Authors:**

Nicola J Robertson (n.robertson@ucl.ac.uk)
Cornelia F Hagmann (corneliahagmann@london.com)
Dominique Acolet (d.acolet@imperial.ac.uk)
Elizabeth Allen (Elizabeth.Allen@lshtm.ac.uk)
Natasha Nyombi (nyombin@yahoo.com)
Diana Elbourne (Diana.Elbourne@lshtm.ac.uk)
Anthony Costello (cihdcostello@gmail.com)
Ian Jacobs (i.jacobs@ucl.ac.uk)
Margaret Nakakeeto (mnakakeeto@yahoo.co.uk)
Frances Cowan (f.cowan@imperial.ac.uk)

**Version:** 7  **Date:** 2 June 2011

**Author's response to reviews:**

I removed two extra commas only. Otherwise the manuscript is fine.